UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-K/A Amendment No. 1

AA	ΛI	N	ON	E

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 FOR THE FISCAL YEAR ENDED DECEMBER 31, 2006

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 FOR THE TRANSITION PERIOD FROM TO

COMMISSION FILE NUMBER 000-52008

LUNA INNOVATIONS INCORPORATED

(Exact name of Registrant as Specified in its Charter)

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

54-1560050

(I.R.S. Employer Identification Number)

1703 South Jefferson Street, SW, Suite 400

Roanoke, VA 24016

(Address of Principal Executive Offices)

(540) 769-8400

(Registrant's Telephone Number, Including Area Code)

Securities registered pursuant to Section 12(b) of the Act:

Title of Each Class:

Name of Each Exchange on which Registered

Common Stock, par value \$0.001 per share

The NASDAQ Stock Market LLC

Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes \square No \boxtimes Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes \square No \boxtimes

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. \Box

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer. See definition of "accelerated filer and large accelerated filer" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer \square Accelerated filer \square Non-accelerated filer \boxtimes

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \square No \boxtimes

The aggregate market value of the registrant's common stock held by non-affiliates of the registrant on June 30, 2006, based upon the closing sale price of the registrant's common stock on such date as reported by the NASDAQ Global Market (formerly the NASDAQ National Market), was approximately \$40.1 million. Shares of the registrant's common stock held by each officer and director of the registrant, and each entity or person that, to the registrant's knowledge, owned 10% or more of the registrant's outstanding common stock as of June 30, 2006, have been excluded in that such persons or entities may be deemed to be affiliates. This determination regarding affiliate status is not necessarily a conclusive determination for other purposes.

Indicate the number of shares outstanding of each of the registrant's classes of common stock, as of the latest practicable date: As of March 15, 2007, there were 10,002,686 shares of the registrant's common stock outstanding.

DOCUMENTS INCORPORATED BY REFERENCE

Specified portions of the registrant's Proxy Statement with respect to its 2007 annual meeting of stockholders are incorporated by reference in Part III of this annual report on Form 10-K.

EXPLANATORY NOTE

This amendment is filed solely to attach Exhibit 10.21 (Asset Transfer and License Agreement by and between Luna Innovations Incorporated and Coherant, Inc.), which was inadvertently omitted from the initial filing of the report and for which confidential treatment is requested, and to correct the incorporation by reference information for Exhibit 10.22 (Form of Stock Sale Restriction Letter Agreement). No other changes to the Annual Report on Form 10-K of Luna Innovations Incorporated, originally filed on March 30, 2007, have been made.

PART IV

ITEM 15. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

- The following documents are filed as part of this Annual Report on Form 10-K, as amended:
 - (1) Financial Statements. See Index to Consolidated Financial Statements at Item 8 of this Report on Form 10-K.
- (2) All other schedules are omitted as the required information is inapplicable or the information is presented in the Consolidated Financial Statements and notes thereto in Item 8 of Part II of this Annual Report on Form 10-K.
- (3) Exhibits. The exhibits filed as part of this report are listed under "Exhibits" at subsection (b) of this Item 15, as amended:
- (b) Exhibits

EXHIBIT INDEX

Exhibit No.	Exhibit Document
3.1(1)	Amended and Restated Certificate of Incorporation of the Registrant (Exhibit 3.2)
3.2(2)	Amended and Restated Bylaws of the Registrant (Exhibit 3.4)
4.1(3)	Specimen Common Stock certificate of the Registrant (Exhibit 4.1)
4.2(2)	2003 Stock Plan (Exhibit 10.7)
4.3(4)	2006 Equity Incentive Plan (Exhibit 10.9)
4.4(2)	Form of Senior Convertible Promissory Note (Exhibit 4.2)
4.5(2)	Warrant to Purchase 1,047 Shares of Class B Common Stock of Luna Innovations Incorporated, issued on September 30, 2005 (Exhibit 4.3)
4.6(2)	Warrant to Purchase 2,636 Shares of Class B Common Stock of Luna Innovations Incorporated, issued on November 11, 2005 (Exhibit 4.4)
4.7(2)	Warrant to Purchase 2,554 Shares of Class B Common Stock of Luna Innovations Incorporated, issued on September 30, 2005 (Exhibit 4.5)
4.8(2)	Form of Warrant to Purchase Shares of Common Stock of Luna Innovations Incorporated (Exhibit 4.6)
4.9(2)	Form of Stock Option Agreement (Exhibit 4.7)
10.1(2)	Form of Indemnification Agreement for directors and executive officers (Exhibit 10.1)
10.2(5)	Employment Agreement by and between the Company and Kent A. Murphy (Exhibit 10.1)
10.3(6)	Employment Agreement by and between the Company and Dale E. Messick (Exhibit 10.1)
10.4(5)	Employment Agreement by and between the Company and John T. Goehrke (Exhibit 10.2)
10.5(7)	Amended and Restated Employment Agreement by and between the Company and Scott A. Graeff (Exhibit 10.1)
10.6(3)	Amended Loan Agreement, dated as of May 12, 2006, by and between Luna Innovations Incorporated and First National Bank (Exhibit 10.6)
10.7(2)	Amended and Restated Investor Rights Agreement, dated December 30, 2005, by and among Luna Innovations Incorporated, Carilion Health System and certain stockholders (Exhibit 10.8)
10.8(8)	Amended Lease, dated July 20, 2006, by and between Carilion Medical Center and Luna Innovations Incorporated. (Riverside Center, Roanoke, Virginia) (Exhibit 10.1)
10.9(9)	Industrial Lease Agreement, dated March 21, 2006, by and between Luna Innovations Incorporated and the Industrial Development Authority of Montgomery County, Virginia (3157 State Street, Blacksburg, Virginia) (Exhibit 10.27)
10.10(3)	First Amendment to Industrial Lease Agreement, dated May 11, 2006, by and between Luna Innovations Incorporated and the Industrial Development Authority of Montgomery County, Virginia (3150 State Street, Blacksburg, Virginia) (Exhibit 10.34)
10.11(2)	Commercial Lease, dated March 17, 2003, between Canvasback Real Estate & Investments LLC and Luna Innovations Incorporated (705 Dale Avenue, Charlottesville, Virginia) (Exhibit 10.14)
10.12(2)	Full Service Office Lease, dated August 2003, between Hampton R&D Properties, LLC and Luna Innovations Incorporated (130 Research Drive, Hampton, Virginia) (Exhibit 10.15)

Exhibit No.	Exhibit Document				
10.13(2)	Lease, effective as of January 1, 2005, between the Industrial Development Authority of Danville and Luna Innovations Incorporated (521 Bridge Street, Danville, Virginia) (Exhibit 10.17)				
10.14(2)	Grant Agreement, dated March 25, 2004, by and between the City of Danville, Virginia, and Luna Innovations Incorporated (Exhibit 10.21)				
10.15(3)	License Agreement No. DN-982, dated June 10, 2002, by and between the National Aeronautics and Space Administration (NASA) and Luna Innovations Incorporated; Modification No. 1 to License Agreement No. DN-982, dated January 23, 2006, by and between NASA and Luna Innovations Incorporated (Exhibit 10.22)				
10.16(3)	License Agreement No. DN-951, dated December 20, 2000, by and between NASA and Luna Technologies, Inc. (Exhibit 10.23)				
10.17(3)	License Agreement No. DE-384, dated October 28, 2004, by and between NASA and Luna Technologies, Inc. (Exhibit 10.24)				
10.18(3)	Fiber Optic Patent License, dated September 22, 2003, by and between United Technologies Corporation and Luna Innovations Incorporated (Exhibit 10.25)				
10.19(3)	Amended and Restated License Agreement, dated March 19, 2004, by and between Virginia Tech Intellectual Properties, Inc. and Luna Innovations Incorporated (Exhibit 10.26)				
10.20(10)	Co-Operation Agreement, dated August 10, 2006, by and between Luna Technologies, Inc. and Acterna France SAS (Exhibit 10.6)				
10.21†	Asset Transfer and License Agreement by and between Luna Innovations Incorporated and Coherent, Inc.				
10.22(9)	Form of Stock Sale Restriction Letter Agreement				
10.23(11)	Amended and Restated Stock Sale Restriction Agreement by and between Luna Innovations Incorporated and Kent A. Murphy, dated as of January 23, 2007. (Exhibit 10.1)				
10.24(11)	Amended and Restated Stock Sale Restriction Agreement by and between Luna Innovations Incorporated and Dale E. Messick, dated as of January 23, 2007. (Exhibit 10.2)				
10.25(11)	Amended and Restated Stock Sale Restriction Agreement by and between Luna Innovations Incorporated and Scott A. Graeff, dated as of January 23, 2007. (Exhibit 10.3)				
10.26(11)	Amended and Restated Stock Sale Restriction Agreement by and between Luna Innovations Incorporated and Robert P. Lenk, dated as of January 23, 2007. (Exhibit 10.4)				
10.27(11)	Amended and Restated Stock Sale Restriction Agreement by and between Luna Innovations Incorporated and Scott A. Meller, dated as of January 23, 2007. (Exhibit 10.5)				
10.28(11)	Amended and Restated Stock Sale Restriction Agreement by and between Luna Innovations Incorporated and Michael F. Gunther, dated as of January 23, 2007. (Exhibit 10.6)				
21.1*	List of Subsidiaries				
23.1*	Consent of Grant Thornton LLP, Independent Registered Public Accounting Firm				
24.1*	Power of Attorney (see signature page)				

31.1*

Certification of the Chief Executive Officer pursuant to Rule 13a-14(a) and Rule 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.

Exhibit No.	Exhibit Document
31.2*	Certification of the Chief Financial Officer pursuant to Rule 13a-14(a) and Rule 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1*	Certification of Chief Executive Officer Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2*	Certification of Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

- (1) Incorporated by reference to the exhibit to the Registrant's Current Report on Form 8-K, Commission File No. 000-52008, dated June 2, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form 8-K.
- (2) Incorporated by reference to the exhibit to the Registrant's Registration Statement on Form S-1, Commission File No. 333-131764, filed on February 10, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form S-1.
- (3) Incorporated by reference to the exhibit to Amendment No. 5 of the Registrant's Registration Statement on Form S-1, Commission File No. 333-131764, filed on May 19, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form S-1.
- (4) Incorporated by reference to the exhibit to Amendment No. 3 of the Registrant's Registration Statement on Form S-1, Commission File No. 333-131764, filed on April 28, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form S-1.
- (5) Incorporated by reference to the exhibit to the Registrant's Current Report on Form 8-K, Commission File No. 000-52008, dated July 14, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form 8-K.
- (6) Incorporated by reference to the exhibit to the Registrant's Current Report on Form 8-K, Commission File No. 000-52008, dated August 29, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form 8-K.
- (7) Incorporated by reference to the exhibit to the Registrant's Current Report on Form 8-K, Commission File No. 000-52008, dated December 20, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form 8-K.
- (8) Incorporated by reference to the exhibit to the Registrant's Current Report on Form 8-K, Commission File No. 000-52008, dated July 20, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form 8-K.
- (9) Incorporated by reference to the exhibit to Amendment No. 2 of the Registrant's Registration Statement on Form S-1, Commission File No. 333-131764, filed on April 10, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form S-1.
- (10) Incorporated by reference to the exhibit to the Registrant's Quarterly Report on Form 10-Q, Commission File No. 000-52008, dated November 13, 2006. The number given in parentheses indicates the corresponding exhibit number in such Form 10-Q.
- (11) Incorporated by reference to the exhibit to the Registrant's Current Report on Form 8-K, Commission File No. 000-52008, dated January 23, 2007. The number given in parentheses indicates the corresponding exhibit number in such Form 8-K.
- * Previously filed.
- † Confidential treatment is requested.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

LUNA INNOVATIONS INCORPORATED

By: /s/ DALE E. MESSICK

Dale E. Messick Chief Financial Officer

April 5, 2007

CONFIDENTIAL TREATMENT REQUESTED: Certain portions of this document have been omitted pursuant to a request for confidential treatment and, where applicable, have been marked with an asterisk ("[****]") to denote where omissions have been made. The confidential material has been filed separately with the Securities and Exchange Commission.

EXECUTION VERSION

ASSET TRANSFER AND LICENSE AGREEMENT

by and between

Coherent, Inc.

and

Luna Innovations Incorporated

Dated December 12, 2006

TABLE OF CONTENTS

1	CONCEDITION AND DEFINITIONS	Page
1.	CONSTRUCTION AND DEFINITIONS 1.1 Definitions	1 1
	1.1 Definitions 1.2 Construction	6
2.		6
	2.1 <u>Transferred Assets</u>	6
	2.2 Excluded Assets	7
	2.3 <u>Transfer of Title to the Assets; Risk of Loss</u>	7
	2.4 Assumption of Liabilities	7
	2.5 <u>Delivery</u>	7
	2.6 Consideration	8
	2.7 Allocation of Purchase Price	8
	2.8 <u>Transfer Taxes</u>	8
3.		9
	3.1 <u>Technology License</u>	9
	3.2 Patent License	10
	3.3 <u>License Limitations</u>	10
	3.4 <u>Transferability</u>	10
	3.5 <u>No Warranties</u>	10
4.	TECHNICAL TRAINING AND ASSISTANCE	11
	4.1 <u>Training</u>	11
	4.2 <u>Training and Assistance Costs</u>	11
	4.3 Additional Training and Assistance	12
	4.4 <u>Assistance with Suppliers</u>	12
	4.5 <u>No Warranties</u>	12
	4.6 Additional Equipment	12
5.	ROYALTIES	13
	5.1 Royalties	13
	5.2 <u>Minimum Annual Royalties</u>	13
	5.3 Royalty Credit	13
	5.4 Reports	14
	5.5 Records	14
	5.6 Payments	14
	5.7 <u>Late Payments</u>	14
	5.8 <u>Currency Conversion</u>	14
6.	SALES TO COHERENT	15
	6.1 Supply Commitment	15
	6.2 <u>Supply Terms</u>	15

7.1 Representations and Warranties of Coherent 7.2 Representations and Warranties of Luna 15	7.	REPR	ESENTATIONS AND WARRANTIES	15
7.2 Representations and Warranties of Luna 18 8. EXESTITIONS CONCURRENT 18 8. I Effectiveness of Luna's Obligations 19 9. I HER COVENANTS AND AGREEMENTS 19 9. I Additional Documents and Further Assurance 20 9. I Additional Documents and Further Assurance 20 10. I Express 21 10. I Assurance 21 10. I I Table Liability 21 10. I Table Liability 21 11. I Termination 21 11. I Termination 22 12. I Table Effect of Termination 22 12. I Table Effect of Termination 23 <		7.1	Representations and Warranties of Coherent	
8.1 Effectiveness of Coherent's Obligations 18 8.2 Effectiveness of Coherent's Obligations 19 9.1 Additional Documents and Further Assurance 19 9.2 Confidential Information 19 9.3 Issurance 20 9.5 Cooperation in Litigation 20 10. IMITATIONS OF LIABILITY 21 11. Intel Liability 21 12. 10.2 EXCLUSION OF DAMAGES 21 12. 11. Termination 21 11. Item AND TERMINATION 21 12. 11. Itemination 21 12. Interest of Termination 22 12. Interest of Termination 22 12. Interest of Segue 23 12. Automys' Fees 24 <td></td> <td>7.2</td> <td></td> <td></td>		7.2		
8.1 Effectiveness of Coherent's Obligations 18 8.2 Effectiveness of Coherent's Obligations 19 9.1 Additional Documents and Further Assurance 19 9.2 Confidential Information 19 9.3 Issurance 20 9.5 Cooperation in Litigation 20 10. IMITATIONS OF LIABILITY 21 11. 10.1 Intel Liability 21 12. 10.2 EXCLUSION OF DAMAGES 21 13. 11. Termination 21 14. ITEM AND TERMINATION 21 12. 11.1 Termination 22 12. ITEMINATION 22 12. 11.4 Effect of Termination 22 12. ITEMINATION 22 12. IAGNIC of Termination 22 12. IAGNIC of Termination 22 12. IAGNIC of Termination 22 12. IAGNIC of Security 23 12.	8.	COND	DITIONS CONCURRENT	18
8.2 Effectiveness of Coheren's Obligations 19 9. OTHER COVENATS AND AGREEMENTS 19 9.1 Additional Documents and Further Assurance 19 9.2 Confidential Information 10 9.3 Insurance 20 9.4 Cooperation in Litigation 20 10. Discoperation In Litigation 21 10. Interpretation 21 10.1 Total Liability 21 10.2 EXCLUSION OF DAMAGES 21 10.3 Fallure of Escential Purpose 21 11. Term 21 11.1 Term 21 11.2 Termination 21 12.1 Permination 22 12.2 Percetaljeure 22 12.3 Porce Majeure 22 12.4 Autorneys' Fees 23 12.5 Okices 23 12.6 Coverning Law 23 12.7 Districtive Relief 24 12.1 Districtive Relief 24 12.2 Savignment 24 12.1 Autorneys' Fees 25 12.2 Autorneys' Fees 23 12.2 Autorneys' Fees 23 12.3 Experts 24 <t< td=""><td></td><td></td><td></td><td></td></t<>				
9.1 Additional Documents and Further Assurance 19 9.2 Confidential Information 19 9.3 Insurance 20 9.5 Cooperation in Litigation 20 10. 1.IMITATIONS OF LIABILITY 21 10.1 Intal Liability 21 10.2 EXCLUSION OF DAMAGES 21 10.3 Tailure of Essential Purpose 21 11. Term 21 11.1 Termination 21 11.2 Termination 22 12. General Termination 22 12. General Termination 22 12. Fefect of Termination 22 12. General Termination 22 12. Force Majeure 22 12. So Agency 22 12. So Agency 22 12. Soverability 23 12. Injunctive Relief 24 12. Amendments 24 12. Amendments 25 12. Amendments 25				
9.1 Additional Documents and Further Assurance 19 9.2 Confidential Information 19 9.3 Insurance 20 9.5 Cooperation in Litigation 20 10. 1.IMITATIONS OF LIABILITY 21 10.1 Intal Liability 21 10.2 EXCLUSION OF DAMAGES 21 10.3 Tailure of Essential Purpose 21 11. Term 21 11.1 Termination 21 11.2 Termination 22 12. General Termination 22 12. General Termination 22 12. Fefect of Termination 22 12. General Termination 22 12. Force Majeure 22 12. So Agency 22 12. So Agency 22 12. Soverability 23 12. Injunctive Relief 24 12. Amendments 24 12. Amendments 25 12. Amendments 25	a	OTHE	PR COVENANTS AND ACREEMENTS	10
9.2 Solidential Information 19 9.3 Insurance 20 9.4 Cooperation in Litigation 20 9.5 Cooperation Regarding Training Obligations 21 10. LITITATIONS OF LIABILITY 21 10.1 Dial Liability 21 10.2 EXCLUSION OF DAMAGES 21 10.3 Failure of Essential Purpose 21 11. TERM AND TERMINATION 21 11.1 Perm 21 11.2 Fermination 21 11.4 Fere 21 12.1 A Solice of Fermination 22 12.2 GENEX 22 12.3 Force Majeure 22 12.4 A Moorneys' Fees 23 12.5 Oxerability 23 12.6 Geverning Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.1 Amendments 24 12.1 Amendments 25 12.1 A Monthey Spenent 25 12.1 A Monthey Spenent 26 12.1 A Mendments 25 12.1 A Mendments 25 12.1 A Mendments	۶.			
9.3 Issurance 20 9.4 Cooperation in Litigation 20 9.5 Cooperation Regarding Training Obligations 21 10. LIMITIONS OF LIABILITY 21 11. 2 Issurance 21 10.3 Failur of Essential Purpose 21 11. TERM AND TERMINATION 21 11.1 Termination 21 11.2 Termination 21 11.3 Notice of Termination 22 12. GENERAL 22 12. Josephs 22 12. No Agency 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Oxices 23 12.5 Notices 23 12.6 Oxices (Severning Law) 23 12.7 Injunctive Relief 24 12.1 Severability 24 12.1 Amendments 24 12.1 Amendments 25 12.1 Amendments 25 12.1 Acknowledgement 25 12.1 Acknowledgement 25		-		
9.4 Cooperation in Litigation (20) 9.5 Cooperation Regarding Training Obligations 10. LIMITONS OF LIABILITY 21 10.1 Total Liability 21 10.2 EXCLUSION OF DAMAGES 21 10.1 TERMINATION 21 11.1 Term 21 11.2 Termination 21 11.3 Oxice of Termination 22 11.4 Effect of Termination 22 12.2 Porce Majeure 23 12.3 Porce Majeure 23 12.4 Autoricys' Fees 23 12.5 Assignment 24 12.1				
9.5 Cooperation Regarding Training Obligations 21 10. LIM™TIONS OF LIABILITY 21 10.1 Total Liability 21 10.2 EXCLUSION OF DAMAGES 21 10.3 Failure of Essential Purpose 21 11. TERM AND TERMINATION 21 11.1 Term 21 11.2 Permination 21 11.3 Notice of Termination 22 12. Hitz 1. Effect of Termination 22 12. GENEAL 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.10 Sasignment 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.17 Acknowledgement 25				
10. LIMITATIONS OF LIABILITY 21 10.1				
10.1 Total Liability 221 232 232 233 234 235	10			
10.2 EXCLUSION OF DAMAGES 21 10.3 Failure of Essential Purpose 21 21 21 21 21 21 21 2	10.			
10.3 Failure of Essential Purpose 21 11. TERM AND TERMINATION 21 11.1 Termination 21 11.2 Termination 22 11.3 Notice of Termination 22 12. GENERAL 22 12.1 Force Majeure 22 12.2 Pos Agency 22 12.3 Pos Agency 22 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.11 Entire Agreement 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 No Third Party Rights 25 12.15 Acknowledgement 25				
11. TERM AND TERMINATION 21 11.1 Term 21 11.2 Termination 21 11.3 Notice of Termination 22 11.4 Effect of Termination 22 12. GENERAL 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
11.1 Term 21 11.2 Termination 21 11.3 Notice of Termination 22 11.4 Effect of Termination 22 12. GENERAL 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
11.2 Termination 21 11.3 Notice of Termination 22 11.4 Effect of Termination 22 12. GENERAL 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.1 Entire Agreement 24 12.1 Amendments 25 12.1 No Third Party Rights 25 12.1 No Third Party Rights 25 12.1 Public Announcements 25 12.1 Acknowledgement 25	11.			
11.3 Notice of Termination 22 11.4 Effect of Termination 22 12. GENERAL 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.10 Severability 24 12.11 Euritre Agreement 24 12.12 Amendments 25 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
11.4 Effect of Termination 22 12. GENETAL 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.10 Sesignment 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12. GENERAL 22 12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.1 Force Majeure 22 12.2 No Agency 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.2 No Agency. 22 12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25	12.			
12.3 Fees and Expenses 23 12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25			·	
12.4 Attorneys' Fees 23 12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.5 Notices 23 12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.6 Governing Law 23 12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.7 Injunctive Relief 24 12.8 Waiver 24 12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.8 Waiver 24 12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.9 Assignment 24 12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.10 Severability 24 12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.11 Entire Agreement 24 12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.12 Amendments 24 12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25			<u>y</u>	
12.13 Exports 25 12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.14 No Third Party Rights 25 12.15 Exhibits and Schedules 25 12.16 Public Announcements 25 12.17 Acknowledgement 25				
12.15Exhibits and Schedules2512.16Public Announcements2512.17Acknowledgement25				
12.16Public Announcements2512.17Acknowledgement25				
12.17 Acknowledgement 25				

Schedules:

Schedule 1.1(e) Schedule 1.1(bb) Clean Room Schedule Licensed Patents

Bill of Material for Transferred Lasers

Schedule 1.1(tt) Schedule 2.1 Transferred Assets Excluded Assets
Prior License Restrictions
Individuals Schedule 2.2 Schedule 3.3

Schedule 7.1(i)

ASSET TRANSFER AND LICENSE AGREEMENT

This Asset Transfer and License Agreement (this "**Agreement**") is entered into as of December 12, 2006 (the "**Effective Date**") by and between Coherent, Inc., a Delaware corporation with a place of business at 5100 Patrick Henry Drive, Santa Clara, California 95054 ("**Coherent**") and Luna Innovations Incorporated, a Delaware corporation with offices at 1703 S. Jefferson Street, SW, Suite 400, Roanoke, Virginia 24014 ("**Luna**") (each, a "**Party**;" together, the "**Parties**").

RECITALS

- A. Coherent operates a Clean Room (as defined below) where it manufactures certain swept-tunable Lasers (as defined below) (the "Business").
- B. Luna wishes to have its Contract Manufacturer (as defined below) assume the manufacture of the Transferred Lasers (as defined below) using certain of the equipment and technology currently employed by Coherent in the Business.
- C. After visiting and performing a detailed technical review of the Clean Room and other due diligence by Luna, Luna has determined the specific items of equipment it wishes to purchase from Coherent, and acknowledges that the equipment it has selected may not be all of the equipment and technology used by Coherent to manufacture the Lasers.
 - D. Coherent is willing to sell to Luna, and Luna is willing to purchase the Transferred Assets (as defined below).
- E. Coherent is willing to grant to Luna, and Luna is willing to accept, a royalty bearing license to manufacture and sell Lasers based upon the Coherent technology.
 - F. Coherent wishes to have the right to purchase Lasers from Luna.

NOW, THEREFORE, in consideration of the foregoing and of the mutual promises contained in this Agreement, the Parties hereby agree as follows:

1. CONSTRUCTION AND DEFINITIONS

- 1.1 **<u>Definitions</u>**. The following capitalized terms have the meanings set forth below:
- (a) "Affiliate" means any entity that controls, is controlled by or is under common control with a Party. An entity shall be regarded as in "control" of another entity, if it owns or possesses, directly or indirectly: (i) voting shares or other securities, representing more than fifty percent (50%) of the outstanding shares or securities entitled to vote for the election of the board of directors or similar managing authority of such controlled entity; or (ii) if such controlling entity does not have voting shares or other securities, more than fifty percent (50%) of the ownership interest that represents the right to make decisions, including the election of directors, for such controlled entity.

- (b) "Assumed Liabilities" has the meaning set forth in Section 2.4.
- (c) "Business Day" means any day other than a Saturday, Sunday or a legal holiday under the federal laws of the United States or any other day on which banking institutions located in California are authorized or required by law or other governmental action to close.
- (d) "Change of Control" means a transaction or series of related transactions that would directly or indirectly: (i) result in or have the effect of a third party obtaining legal or beneficial ownership of more than fifty percent (50%) of the voting shares (or other voting interests) of a Party (even if the Party is the surviving entity, such as in the case of a reverse triangular merger); or (ii) result in the sale, transfer, assignment, mortgage, exclusive license or other disposition of all or substantially all of a Party's assets.
- (e) "Clean Room Schedule" means Schedule 1.1(e) hereto, the list of material equipment in the Clean Room as of the date hereof that is used by Coherent in the manufacture of the Transferred Lasers in the Clean Room, but excluding commercially available portable maintenance, calibration and test equipment and commercially available computers and network equipment.
 - (f) "Clean Room" means the Coherent Clean Room located at 1870 Lundy Avenue, San Jose, California.
- (g) "Coherent Lasers" means the swept-tunable lasers, including the Transferred Lasers, manufactured by Coherent prior to the Effective Date using the Transferred Assets in the Clean Room.
 - (h) "Confidential Information" has the meaning set forth in Section 9.2.
- (i) "Contract Manufacturer" means an entity that will use the Transferred Assets to manufacture for, and supply lasers to, Luna and, if requested, Coherent, and any replacement of, or successor to, such contract manufacturer. The Contract Manufacturer shall initially be Avo Photonics.
- (j) "Contract Year" means the twelve (12)-month period commencing on January 1, 2007, and each separate successive twelve (12) month period thereafter during the Term of this Agreement.
 - (k) "Credit Amount" has the meaning set forth in Section 5.3.
- (l) "<u>Derivative Work</u>" has the meaning ascribed to it under the United States Copyright Law, Title 17 U.S.C. Sec. 101 et. seq., as the same may be amended from time to time.

- (m) "Disclosing Party" has the meaning set forth in Section 9.2.
- (n) "<u>Documentation</u>" means all documents in existence as of the Effective Date and necessary to the manufacture, use and testing of Transferred Lasers or the operation of the Transferred Equipment, including but not limited to the design and manufacturing documents (such as bill of materials, build instructions and test reports and vendor lists (but only to the extent Coherent is permitted by applicable laws, regulations and agreements to provide the same to Luna without restriction).
 - (o) "Excluded Assets" has the meaning set forth in Section 2.2.
 - (p) "Excluded Liabilities" has the meaning set forth in Section 2.4.
- (q) "Fields of Use" means (i) for Lasers that operate from 1250 nm to 1700 nm, use in any application, and (ii) for Lasers that operate over any wavelength range other than the wavelengths set forth in the foregoing clause (i), for use only in applications in the fields of telecommunications, telecommunications test and/or coherent optical frequency domain reflectometry.
 - (r) "First Payment" has the meaning set forth in Section 2.6.
- (s) "Full Payment Date" means the date upon which the full Purchase Price has been paid by Luna, subject to Luna's right to offset, in accordance with Section 2.6.
- (t) "GAAP" means generally accepted accounting principals as may be modified from time to time by the Securities and Exchange Commission, the Public Company Accounting Oversight Board and other recognized authorities.
- (u) "Governmental Authority" means any U.S. or foreign, national, federal, state, municipal or local or other government, governmental, regulatory or administrative authority, agency or commission of any court, tribunal, or judicial or arbitral body.
- (v) "Intellectual Property Rights" means the rights associated with the following: (i) all United States and foreign patents and applications, including continuation, divisional and continuation-in-part applications thereof as well as any reissues and reexaminations thereof ("Patents"); (ii) all trade-secret rights and all other rights in or to confidential business or technical information ("Trade Secrets"); (iii) all copyrights, copyright registrations and applications therefor and all other rights corresponding thereto throughout the world ("Copyrights"); (iv) trademarks, service marks, trade dress rights and similar designation of origin and rights therein ("Marks"); and (v) any similar, corresponding or equivalent rights to any of the foregoing any where in the world. Intellectual Property Rights specifically excludes contractual rights, including license grants.
- (w) "<u>Inventory</u>" means all inventories of (i) raw and in-process materials, parts and supplies used in the manufacture of Lasers that constitute the Transferred Lasers listed on Schedule 1.1(tt), and (ii) the final, completed inventory of Lasers that constitute the Transferred Lasers listed on Schedule 2.1.

- (x) "**Knowledge**" has the meaning set forth in Section 7.1.
- (y) "Lasers" means swept-tunable lasers generally made by Coherent or Luna using the Transferred Technology in whole or part.
- (z) "Licenses" means the Licensed IPR, the Licensed Patents and any other licenses granted by Coherent under this Agreement.
- (aa) "<u>Licensed IPR</u>" means any Intellectual Property Rights (other than Patents and Marks) owned or licensed by Coherent as of the Effective Date that are embodied by the Transferred Technology delivered under this Agreement and necessary to the manufacture, use and testing of the Transferred Lasers.
- (bb) "<u>Licensed Patents</u>" means those Patents listed on Schedule 1.1(bb), which represent the Patents owned, licensed or controlled by Coherent and necessary to make and have made Licensed Products and to sell, offer for sale, import and use any such Licensed Products in the Field of Use.
- (cc) "Licensed Product" means any device or product that: (i) embodies, comprises or incorporates Lasers; (ii) is made or developed using any Licensed IPR; (iii) is made, had made, used, offered for sale, sold or imported, by Luna pursuant to the licenses granted under this Agreement; or (iv) is covered by a Valid Claim of the Licensed Patents.
- (dd) "Lien" means, with respect to any asset, any mortgage, lien, pledge, charge, security interest, claim, restriction or other encumbrance or defect of title of any nature whatsoever in respect of such asset.
 - (ee) "Luna Technical Personnel" has the meaning set forth in Section 4.1.
 - (ff) "Minimum Royalties" has the meaning set forth in Section 5.2.
- (gg) "Net Revenues" means the greater of the total amount invoiced or received upon Sales of Licensed Products by Luna and its Affiliates to non-Affiliate third parties in arms-length transactions, less the following reasonable and customary deductions if charged separately on the invoice and paid by the customer: (i) rebates, refunds and credits (including amounts repaid by reason of returns, billing errors, retroactive price adjustments and the like), and cash and trade discounts, in each case to the extent actually given; and (ii) insurance, transportation and other delivery charges paid by the seller, and (iii) sales taxes, duties and other governmental charges (including value added tax), but excluding what is commonly known as income taxes. Net Revenues shall not include amounts invoiced or received on Sales of Licensed Products among Luna and its Affiliates for resale, provided that in such case, Net Revenues shall include the amounts invoiced or received on subsequent re-Sales of such Licensed Products to non-Affiliate third parties in arms-length transactions. If any Licensed Products are Sold in whole or in part in exchange for consideration other than cash, Net Revenues shall include the fair market value of such Licensed Products. Further, if Licensed Products are Sold for less than their fair market value (for example, as part of "bundle" of products or as a "loss leader"), such Licensed Products shall be considered for Royalty purposes to have been Sold at Licensee's or its

Affiliate's (as applicable) average, bona fide list price for such Licensed Product to end-user customers over the 12-month period immediately preceding such Sale (such average to be weighted to account for the number of such Licensed Product units Sold while each list price was in effect during such 12-month period).

- (hh) "Permitted Liens" has the meaning set forth in Section 7.1.
- (ii) "Purchase Price" has the meaning set forth in Section 2.6.
- (jj) "**Receiving Party**" has the meaning set forth in Section 9.2.
- (kk) "Royalty" has the meaning set forth in Section 5.1.
- (ll) "Sale" means, for purposes of determining Net Revenues payable under this Agreement, any sale, lease, transaction or other disposition in which a Licensed Product is exchanged for value and Luna recognizes revenue in accordance with GAAP, provided, however, that the provision by Luna of any Laser to a third party for use by such third party solely for use in a research or development activity with Luna shall not be considered a Sale hereunder, even if Luna receives research or development funds from such third party.
- (mm) "Taxe" or "Taxes" means all federal, state and local taxes (including, without limitation, income, profit, franchise, sales, use, real property, personal property, ad valorem, excise, employment, social security and wage withholding taxes) and installments of estimated taxes, assessments, deficiencies, levies, imports, duties, license fees, registration fees, withholdings, or other similar charges of every kind, character or description (domestic or foreign) imposed by any Governmental Authorities responsible for the imposition of any such Tax (a "Tax Authority"), and any interest, penalties or additions to tax imposed thereon or in connection therewith.
- (nn) "<u>Technology</u>." means all tangible embodiments, whether in electronic, written or other media, of all technology, including all techniques, design rules, inventions (whether or not patented or patentable), algorithms, routines, software, databases, processes, prototypes, and devices.
- (oo) "Training Completion Date" means the date on which the requirements related to the Training Obligations set forth in Section 4.1 have been satisfied, which date shall be no later than December 31, 2006.
 - (pp) "Training Obligations" has the meaning set forth in Section 4.1.
 - (qq) "Training Period" has the meaning set forth in Section 4.1.
 - (rr) "Transferred Assets" has the meaning set forth in Section 2.1.
 - (ss) "Transferred Equipment" means all equipment, machinery, tools and other tangible personal property set forth on Schedule 2.1 hereto.

- (tt) "<u>Transferred Lasers</u>" means the particular product form of swept-tunable Lasers that as of the date hereof are being manufactured by Coherent in the Clean Room and supplied to Luna and the bill of materials for which product is listed on Schedule 1.1(tt).
- (uu) "<u>Transferred Software</u>" means the software authored by, or on behalf of, or licensed by Coherent and used in the operation of the Clean Room and manufacture of the Transferred Lasers, including the control of the Transferred Equipment. The Transferred Software will include both source code and object code to the extent available.
- (vv) "<u>Transferred Technology</u>" means all the Transferred Equipment, the Documentation and other Technology owned and transferable by Coherent that is included within the Transferred Assets or disclosed by Coherent to Luna hereunder.
 - (ww) "Transfer Taxes" has the meaning set forth in Section 2.8.
- (xx) "Valid Claim" means a claim of any Patent which has not been declared unpatentable by the applicable national or regional patent authority or invalid by a court of competent jurisdiction in an unappealed and unappealable decision.

1.2 Construction.

(a) For purposes of this Agreement, whenever the context requires: the singular number will include the plural, and vice versa; the masculine gender will include the feminine and neuter genders; the feminine gender will include the masculine and neuter genders; and the neuter gender will include the masculine and feminine genders. The Parties hereto agree that they have had the opportunity to discuss this Agreement with and obtain advice from their legal counsel, have had sufficient time to, and have carefully read and fully understand all the provisions of this Agreement, and are knowingly and voluntarily entering into this Agreement. Therefore, the Parties waive the application of any law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the Party drafting such agreement or document. As used in this Agreement, the words "include" and "including," and variations thereof, will not be deemed to be terms of limitation, but rather will be deemed to be followed by the words "without limitation." Except as otherwise indicated, all references in this Agreement to "Sections" and "Exhibits" are intended to refer to Sections of this Agreement and Exhibits to this Agreement. The headings in this Agreement are for convenience of reference only, will not be deemed to be a part of this Agreement, and will not be referred to in connection with the construction or interpretation of this Agreement.

2. PURCHASE AND SALE

- 2.1 <u>Transferred Assets</u>. On the terms and subject to the conditions set forth in this Agreement, Coherent hereby sells, transfers, and assigns to Luna, and Luna purchases, assumes, and acquires from Coherent, all rights of Coherent (other than Intellectual Property Rights which are subject to the provisions of Article 3) in the following (collectively, the "<u>Transferred Assets</u>"):
 - (i) the Transferred Equipment and other tangible assets listed on Schedule 2.1;

- (ii) the Inventory listed on Schedule 1.1(tt);
- (iii) a copy of the Transferred Software; and
- (iv) a copy of the Documentation.

EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES SET FORTH IN ARTICLE 7, ALL TRANSFERRED ASSETS ARE PROVIDED "AS IS," "WHERE IS" AND WITHOUT WARRANTY OF ANY KIND.

- 2.2 Excluded Assets. Notwithstanding anything to the contrary set forth in this Agreement, but subject to Section 4.6, any assets that are not Transferred Assets are retained by Coherent and will not be sold, transferred or assigned to Luna under this Agreement except as otherwise agreed pursuant to the Agreement (the "Excluded Assets"). Without limitation the Excluded Assets include: (a) any items not otherwise included on Schedule 2.1, and (b) any assets otherwise listed on Schedule 2.2.
- 2.3 <u>Transfer of Title to the Assets</u>; <u>Risk of Loss</u>. Coherent shall sell, convey, transfer and assign the Transferred Assets to Luna as of the Effective Date by means of bills of sale, assignments, endorsements, certificates and such other instruments of transfer as shall be necessary and appropriate to vest title to the Transferred Assets, free and clear of any liens, mortgages, pledges and similar encumbrances, in Luna as and to the extent provided in this Agreement or in the Schedules hereto. Subject to the terms and conditions of this Agreement, risk of loss with respect to the Transferred Assets will pass to Luna upon physical delivery of the Transferred Assets to and acceptance of delivery by Luna to occur at the Clean Room, provided that Coherent shall add Luna as an additional insured party to its property insurance policy covering the Transferred Assets effective as of the Effective Date, which coverage shall end not less than five (5) Business Days following the Training Completion Date.
- 2.4 <u>Assumption of Liabilities</u>. Effective as of the Effective Date, Luna assumes and is responsible for all obligations and liabilities solely relating to, arising out of, or resulting from the Transferred Assets on or after the Effective Date, or to the exercise of the license rights granted under this Agreement (collectively, the "<u>Assumed Liabilities</u>"). Coherent, without any responsibility or liability of or recourse to Luna or any of Luna's directors, stockholders, officers, employees, agents, consultants, attorneys, representatives, affiliates, successors or assigns, shall absolutely and irrevocably retain all liabilities (including Taxes) that existed, related to or arose prior to the Effective Date including those relating to Coherent, the Business or the Transferred Assets (the "<u>Excluded Liabilities</u>").

2.5 Delivery.

- (a) Upon the Effective Date, Coherent shall make available to Luna all of the Transferred Assets, Ex Works (Incoterms 2000) the Clean Room or at such other Coherent location as the Parties may mutually agree in writing.
 - (b) Luna shall remove the Transferred Assets from the Clean Room no later than five (5) Business Days after the Training Completion Date.

- (c) Luna will detach and disconnect the Transferred Asset from the Clean Room and from utilities.
- (d) Luna will be solely responsible for otherwise preparing and packaging the Transferred Assets for shipment; provided that Coherent shall have the right to supervise such activities to assure that no damage is done to the Clean Room and no Excluded Assets are removed from the Clean Room.

2.6 Consideration.

- (a) Luna shall pay to Coherent by wire transfer of immediately available funds to an account designated by Coherent the sum of [****] dollars (\$[****]) (the "**Purchase Price**"), payable as follows:
 - (i) [****] dollars (\$[****]) (the "First Payment") shall be due and payable upon the Effective Date;
 - (ii) [****] dollars (\$[****]) shall be due and payable on the one (1)-year anniversary of the Effective Date; and
 - (iii) [****] dollars (\$[****]) shall be due and payable on the two (2)-year anniversary of the Effective Date.
- (b) The obligation of Luna to pay the amounts set forth above shall be subject to the rights and obligations of the Parties set forth herein. Luna shall retain a right of offset against its payment obligations under the Agreement in the event of a material breach by Coherent. In the event of Luna's failure to pay the Purchase Price when due, interest shall be due on all overdue amounts in accordance with Section 5.7. Absent a failure to pay the Purchase Price when due, no interest with respect to the Purchase Price is otherwise due or payable by Luna. Notwithstanding anything herein to the contrary, in the event that a complaint is filed in a court of competent jurisdiction requesting an injunction preventing Luna from using, making, having made, importing, selling or offering for sale Licensed Products, Luna may suspend payments under this Section 2.6 until such action is dismissed and until all appeals are otherwise exhausted.
- 2.7 <u>Allocation of Purchase Price</u>. Within ninety (90) days of the Effective Date, Luna shall provide Coherent with an allocation among the Transferred Assets and the Licenses granted hereunder of the Purchase Price, the Royalty and the amount of the Assumed Liabilities to the extent properly taken into account under Section 1060 of the Internal Revenue Code of 1986, as amended, and the regulations promulgated thereunder (the "Allocation"). The Allocation shall be conclusive and binding upon Luna and Coherent for all purposes, and the parties agree that all returns and reports (including Internal Revenue Service ("IRS") Form 8594) and all financial statements shall be prepared in a manner consistent with (and the parties shall not otherwise take a position on a tax return that is inconsistent with) the Allocation unless required by the IRS or any other applicable taxing authority.

2.8 Transfer Taxes.

- (a) The amounts payable by Luna under this Agreement are exclusive of any future sales, property or excise tax, duties, value added tax, tariffs, lending charges or other taxes, fees and charges. Luna and Coherent shall share equally all expenses with respect to all transfer, documentary, sales, use, stamp, registration, value added, gross receipts, conveyance, excise, recording, license, filing and other such taxes and fees (including any penalties and interest) applicable to, imposed upon or arising out of the transactions contemplated hereby including, without limitation, any transfer tax or filing fee relating to the transfer of personal property ("**Transfer Taxes**"), whether now in effect or hereinafter adopted and regardless of which Party such Transfer Taxes are imposed upon.
- (b) Luna will, at its own expense, file all necessary tax returns and other documentation with respect to all Transfer Taxes imposed upon Luna, and, if required by applicable law, Coherent will join in the execution of any such tax returns and other documentation.
- (c) Coherent will, at its own expense, file all necessary tax returns and other documentation with respect to all Transfer Taxes imposed upon Coherent, and, if required by applicable law, Luna will join in the execution of any such tax returns or other documentation.
- (d) In the case of any personal property taxes (or other similar Taxes) attributable to the Transferred Assets for which Taxes are reported on a tax return covering a period commencing before the Effective Date and ending thereafter (a "Straddle Period Tax"), any such Straddle Period Taxes shall be prorated between Coherent and Luna on a per diem basis. The Party required by law to pay any such Straddle Period Tax (the "Paying Party") shall file the tax return related to such Straddle Period Tax within the time period prescribed by law and shall timely pay such Straddle Period Tax. To the extent any such payment exceeds the obligation of the Paying Party hereunder, the Paying Party shall provide the other party (the "Non-Paying Party") with notice of payment, and within 10 days of receipt of such notice of payment, the Non-Paying Party shall reimburse the Paying Party for the Non-Paying Party's share of such Straddle Period Taxes.
- (e) To the extent relevant to the Transferred Assets, each Party shall (i) provide the other with such assistance as may reasonably be required in connection with the preparation of any tax return and the conduct of any audit or other examination by any taxing authority or in connection with judicial or administrative proceedings relating to any liability for Taxes and (ii) retain and provide the other with all records or other information that may be relevant to the preparation of any tax returns, or the conduct of any audit or examination, or other proceeding relating to Taxes. Coherent shall retain all documents, including prior years' tax returns, supporting work schedules and other records or information with respect to all sales, use and employment tax returns and, absent the receipt by Coherent of the relevant tax clearance certificates, shall not destroy or otherwise dispose of any such records for six (6) years after Effective Date without the prior written consent of Luna.

3. LICENSES

3.1 <u>Technology License</u>. Subject to the terms and conditions of this Agreement, Coherent hereby grants to Luna and its Affiliates under all of Coherent's rights in

the Licensed IPR, a perpetual, world-wide, non-exclusive, right and license to fully use the Transferred Assets delivered hereunder in connection with the design, making, using, selling and maintenance of Licensed Products in the Field of Use.

- 3.2 <u>Patent License</u>. Subject to the terms and conditions of this Agreement, Coherent hereby grants to Luna and its Affiliates, under all of Coherent's rights in the Licensed Patents, a limited, world-wide, non-exclusive, license to make and have made Licensed Products and to sell, offer for sale, import and use any such Licensed Products in the Field of Use; *provided*, *however*, that upon the fulfillment of Luna's Royalty obligations under Article 5 hereof, all Licenses granted by Coherent under this Agreement will become fully paid up.
- (a) The foregoing licenses do not grant to Luna or its Affiliates rights with respect to any Licensed Products other than those for which the designs, specifications and working drawings for manufacture either are delivered hereunder as Transferred Technology or originate with Luna or its Affiliates and are owned exclusively or jointly by Luna or its Affiliates, or to which Luna or its Affiliates otherwise has rights.
- (b) Luna or its Affiliates may not sublicense the rights granted to it in this Section 3.2 to any third party; *provided* that nothing in this Section 3.2(b) shall in any way limit the License rights of Luna or its Affiliates to have made for use and sale by Luna, its Affiliates, distributors, or representatives.
- (c) Luna and its Affiliates shall not exercise any rights, including their make and have made rights, in a manner that is intended to or would have the effect of sublicensing the rights licensed to it under this Section 3.2 to any third party, or that would otherwise expand the subject of the foregoing license beyond the Laser Products described in Section 3.2(a).
- 3.3 <u>License Limitations</u>. All Licenses granted by Coherent to Luna and its Affiliates hereunder shall be subject to any and all licenses and other agreements between Coherent and any third party or parties entered into prior to the Effective Date and that are listed and described on Schedule 3.3 herein. The Licenses granted hereunder do not extend to, and Luna and its Affiliates are not granted a license for, the manufacture, use, import or sale of any Laser intended or designed for use outside the Field of Use.
- 3.4 <u>Transferability</u>. The foregoing Licenses under this Article 3 may be transferable by Luna or its Affiliates, directly or indirectly, to any third party (including to a successor, in connection with a Change of Control, by operation of law, or otherwise), without the prior written consent of Coherent.
- 3.5 **No Warranties**. Subject to the representations and warranties in Article 7 hereof, Coherent makes no, and expressly disclaims, any warranty that exercise by Luna or its Affiliates of any of the forgoing Licenses or that their making, using, selling, or importing of Licensed Products or the use of the Transferred Assets will not infringe the Intellectual Rights of any third party.

4. TECHNICAL TRAINING AND ASSISTANCE

4.1 Training.

- (a) In order to enable Luna and the Contract Manufacturer to understand the operation and use of the Transferred Assets, during the eleven (11) Business Day period beginning on December 4, 2006 (the "**Training Period**"), Coherent will provide up to seven (7) technical personnel of Luna and its Contract Manufacturer ("**Luna Technical Personnel**") with access to, and training in, the process it currently uses to manufacture the Transferred Lasers using the Transferred Assets at the Clean Room (the "**Training Obligations**").
- (b) In order to receive such training, Luna and the Contract Manufacturer (at Luna's option) shall send up to seven Luna Technical Personnel to the Clean Room. Such Luna Technical Personnel may observe and become familiar with the Coherent process for manufacturing the Transferred Lasers and the use of the Transferred Assets at the Clean Room. All such technical personnel shall be (i) in the sole and exclusive permanent employ of Luna or the Contract Manufacturer, and (ii) properly trained and competent in their respective technical areas. The number of technical personnel present at the Clean Room at any one time shall be subject to Coherent's prior approval which shall not be unreasonably withheld.
- (c) Coherent shall cooperate with the Luna Technical Personnel sent to the Clean Room and assist them in learning the manufacturing methods and techniques with respect to the Coherent processes used in the Clean Room and the manufacture of Transferred Lasers.
- (d) Coherent will permit Luna Technical Personnel, under the supervision of Coherent, to build at least ten (10) of the Transferred Lasers, of which at least three (3) Transferred Lasers shall be in good working order and capable of being powered and tuned.
 - (e) Coherent shall provide the Luna Technical Personnel with reasonable office space and communications access while at the Clean Room.
- (f) All Luna Technical Personnel shall observe all of Coherent's reasonable employee rules and regulations and abide by any reasonable instructions from Coherent personnel while at the Clean Room.
 - (g) Luna shall be solely responsible for all of all Luna Technical Personnel's travel, lodging and other expenses.
- (h) Luna shall be solely responsible for, and shall indemnify Coherent for any and all material liabilities that Coherent may suffer or incur solely as a result of any act by any Luna Technical Personnel while on the premises of the Clean Room.

4.2 Training and Assistance Costs.

- (a) Up to eleven (11) days of training and assistance to the Luna Technical Personnel as specified in Section 4.1 shall be provided by Coherent to Luna at no-charge except that Luna shall pay for the reasonable cost of materials and supplies needed for such training (including for the manufacture of Transferred Lasers during such training).
- (b) Coherent shall, upon Luna's request, extend the training set forth in Section 4.2(a) by up to an additional five (5) days provided that (i) Luna pays Coherent an additional amount equal to \$[*****] for each full Business Day of such additional training and assistance, and the cost of materials and supplies needed for such training, and (ii) such additional training is completed by December 31, 2006. Such amount shall be paid within thirty (30) days of Coherent delivering an invoice to Luna for such additional training and assistance and for the cost of materials and supplies used.
- 4.3 <u>Additional Training and Assistance</u>. If Luna requires technical training or assistance in addition to that specified under Sections 4.1 and 4.2 above, Coherent may provide such training and assistance, provided that the Parties agree on compensation therefor. In no event shall Coherent be deemed in breach of this Agreement for failure to provide such additional assistance other than as determined by Coherent in its sole discretion.
- 4.4 <u>Assistance with Suppliers</u>. Coherent shall provide Luna with introductions to all existing suppliers of the materials on the current bill of materials for the Lasers, as defined in Schedule 1.1(tt); provided that Luna shall be solely responsible for purchasing materials from such suppliers.
- 4.5 No Warranties. Luna expressly warrants and acknowledges that the making of Lasers using the Transferred Assets will require significant technical skill and expertise on its and its Contract Manufacturer's behalf, that such skill and expertise will not be provided by the training specified herein, and that Luna shall be solely responsible for having or developing such skills and technical expertise. Without limiting Coherent's Training Obligations set forth in Sections 4.1, 4.2 and 4.3, Coherent makes no warranty and expressly disclaims any warranty that Luna or the Contract Manufacturer will be able to successfully make Lasers using the Transferred Assets.

4.6 Additional Equipment.

(a) If during the Training Period, Luna determines that it wishes to acquire one or more items of equipment listed on the Clean Room Schedule but not listed as a Transferred Asset on Schedule 2.1 (and therefore not previously identified by Luna as an item that it wished to acquire under this Agreement), and that item of equipment are necessary to the manufacture of Transferred Lasers by Luna or the Contract Manufacturer following the Effective Date, then upon Luna's request made prior to the sixtieth (60th) day from the date hereof, the Parties shall enter into good faith negotiations on such additional compensation and other terms as Coherent reasonably requires to add that item as a Transferred Asset on Schedule 2.1 hereunder; provided that if the Parties do not agree to such terms within ten (10) days after commencing such negotiations, Coherent shall have no further obligations with respect to such additional items under this Section 4.6.

(b) If within sixty (60) days of the date hereof it is determined that there was one or more items of equipment in the Clean Room that were not on the Clean Room Schedule but which are necessary to the manufacture of Transferred Lasers by Luna or the Contract Manufacturer following the Effective Date, then, as Luna's sole and exclusive remedy for such omission, Coherent shall transfer and sell such equipment to Luna at price equal to the reasonable replacement cost of such equipment.

5. ROYALTIES

5.1 **Royalties**. Luna, in addition to the consideration set forth in Article 2, shall pay to Coherent a royalty (the "**Royalty**") equal to [****] percent ([****]%) of the Net Revenues from the Sale of Licensed Product Sold by Luna or its Affiliates prior to the end of the tenth (10th) Contract Year. In the event that one or more Lasers is included in a product or system Sold by Luna, then for the purposes of the foregoing Royalty calculation Net Revenues shall refer to the Net Revenues resulting from the Sale of such product or system regardless of whether the Laser included therein is separately priced or shipped.

5.2 Minimum Annual Royalties

(a) . Royalty payments under Section 5.1 shall be subject to the following minimum payment schedule (the "Minimum Royalties"). If actual Royalties calculated in accordance with Section 5.1 for a given Contract Year would be less than the Minimum Royalty for such Contract Year, then Luna shall pay to Coherent the Minimum Royalty for such Contract Year. If the Royalties calculated in accordance with Section 5.1 exceed the Minimum Royalties for such Contract Year, only the Royalties calculated pursuant to Section 5.1 shall be paid. The Section 5.1 Royalties and the Minimum Royalties are not additive. The Minimum Royalties payable for each Contract Year are as follows:

Contract Year	Minimum Royalties
1	\$[****]
2	\$ [****]
3	\$ [****]
4	\$ [****]
5	\$ [****]
6-10	\$[****]

(b) In the event (i) Luna, based on legal counsel's opinion, in good faith believes that making, having made, using, selling or importing Licensed Product infringes any patent, or (ii) a complaint is filed in a court of competent jurisdiction requesting an injunction preventing Luna from using, making, having made, importing, selling or offering for sale the Licensed Products, and until such action is dismissed and all appeals are otherwise exhausted, the requirement to pay the Section 5.2(a) Minimum Royalties shall be suspended.

- 5.3 **Royalty Credit**. Notwithstanding the foregoing, if in any year prior to the end of the fifth (5th) Contract Year, Coherent has purchased Lasers from Luna in accordance with Article 6, Luna may deduct from the calculation of Net Revenues used to determine Royalties payable to Coherent an amount equal to the Net Revenues of products sold by Coherent incorporating Lasers (the "**Credit Amount**"); provided that in no event will the Royalties otherwise payable be reduced in accordance with the foregoing to below the Minimum Royalty for such Contract Year.
- 5.4 **Reports**. Within fifteen (15) days after the end of each Contract Year, Coherent shall deliver to Luna a true and accurate report, giving such particulars of the business conducted by Coherent and its Affiliates, if any, during such Contract Year as are pertinent to an account for payments hereunder. Such report shall include at least (a) the number and value of products sold by Coherent incorporated Lasers during the Contract Year, (b) the Credit Amount calculated under Section 5.3 above attributable to such sales. Within thirty (30) days of the receipt of Coherent's report under this Section 5.4, Luna shall deliver to Coherent a true and accurate report, giving such particulars of the business conducted by Luna and its Affiliates, if any, during such Contract Year as are pertinent to an account for payments hereunder. Such report shall include at least (a) the number of Licensed Products Sold during the Contract Year, (b) the calculation of Net Revenues, (c) the Credit Amount for such Contract Year, and (d) the amount of Royalty payable by Luna to Coherent. Concurrently with the delivery of each such report, Luna shall pay to Coherent the total amounts, if any, due to Coherent for the period of such report. If no amounts are due, Luna shall so report.
- 5.5 **Records**. Each Party shall keep complete and accurate books of account and records for the purpose of determining the amounts payable under this Article 5. Such records will be open for inspection and audit during reasonable business hours by a representative or agent of the other Party, but only once every twelve (12) months, for the purpose of verifying the other Party's reports and payments under this Article 5. If an audit determines that one Party has failed to pay Royalties in full when due, such Party shall immediately pay the amount of such Royalties plus interest calculated from the date such Royalties were due in accordance with Section 5.7; provided no such payment shall be due to the extend that the parties dispute the results of such audit. In such case, the parties shall use best efforts to promptly seek dispute resolution of such dispute. In addition, if such audit establishes an underpayment in Royalties exceeding five percent (5%), the Party responsible for such underpayment shall reimburse the other Party for the cost of such audit.
- 5.6 **Payments**. All amounts payable under this Article 5 shall be made by bank-wire transfer in immediately available funds to an account designated by receiving Party.
- 5.7 <u>Late Payments</u>. Any payment due under this Agreement which is not paid on the date due shall bear interest equal to the lower of (i) the maximum amount permitted by applicable law, and (ii) a monthly simple interest rate of 2%, calculated on the number of days such payment is delinquent. This Section 5.7 shall in no way limit any other remedies available to the non-breaching Party for failure to pay amounts when due.
- 5.8 <u>Currency Conversion</u>. If any currency conversion is required in connection with the calculation of amounts payable under this Article 5, such conversion shall be

made using the average spot exchange rate for conversion of the foreign currency into U.S. Dollars, quoted for current transactions reported in *The Wall Street Journal*, for the last ten (10) Business Days of the Contract Year to which such payment pertains.

6. SALES TO COHERENT

6.1 <u>Supply Commitment</u>. During the term of this Agreement and subject to the terms and conditions set forth in this Article 6, Coherent will have the right to order and purchase from Luna, and Luna will use commercially reasonable efforts to cause to be supplied to Coherent, Lasers (in the version as then being manufactured for sale by Luna).

6.2 Supply Terms.

- (a) Subject to the terms and conditions of this Agreement including but not limited to Luna's or the Contract Manufacturer's ability to successfully manufacture the Lasers, Coherent shall have the right to purchase up to [****] Lasers in the first (1st) Contract Year, up to [****] Lasers in the second (2nd) Contract Year, and up to [****] Lasers in each remaining Contract Year through the end of the tenth (10th) Contract Year. In addition to the foregoing, Coherent may purchase from Luna an additional quantity of Lasers in each Contract Year as it may reasonably request (above the limits set forth in this Section 6.2(a)); provided, that such orders do not impair or adversely impact (i) Luna's business, (ii) Luna's customers, or (iii) the business of Luna's Contract Manufacturer; provided further, that Luna's own supply requirements shall have priority over any such orders from Coherent under this Article 6.
- (b) Luna shall use commercially reasonable efforts to sell such Lasers to Coherent at [****] (as defined below) and subject to such other customary terms and conditions to be negotiated in good faith by the Parties. For purposes of this Section 6.2 only, [****]
- (c) Coherent's right to order and purchase Lasers pursuant to this Article 6 shall terminate upon notice (as set forth in Section 12.5) to Coherent that Luna has ceased or caused to be ceased, or otherwise suspended, the manufacture of Lasers.

7. REPRESENTATIONS AND WARRANTIES

- 7.1 Representations and Warranties of Coherent. Coherent hereby represents and warrants to Luna that:
- (a) <u>Organization of Coherent</u>. Coherent is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- (b) <u>Authority</u>. Coherent has all requisite corporate power and authority to enter into this Agreement and to perform its respective obligations and consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary action (corporate or otherwise) on the part of Coherent, and no further action is required on the part of Coherent to authorize the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby. This Agreement has been duly executed and delivered by Coherent and is valid and enforceable against Coherent in accordance with its terms.

- (c) No Conflict. Neither the execution and delivery of this Agreement, nor the consummation of the transactions contemplated hereby, conflicts or will conflict with, or result in any violation of, or default under (with or without notice or lapse of time, or both), or give rise to a conflict under (i) any provision of Coherent's charter documents or bylaws, each as currently in effect, (ii) any contract to which Coherent or any of its properties or assets (including, but not limited to, the Transferred Assets or Licensed Patents) are subject, or (iii) any judgment, order, writ, injunction, decree, award, statute, law, ordinance, rule or regulation applicable to Coherent or its properties or assets (including, but not limited to, the Transferred Assets or the Licensed Patents). No authorization, approval or consent of, and no registration or filing with, any governmental or regulatory official, body or authority is required in connection with the execution, delivery or performance of this Agreement by Coherent.
- (d) <u>Absence of Undisclosed Liabilities</u>. There are no liabilities or obligations affecting the Transferred Assets as of the date hereof. For purposes of this Agreement, the term liabilities shall include, without limitation, any direct or indirect indebtedness, guaranty, endorsement, claim, loss, damage, deficiency, cost, expense, obligation or responsibility, fixed or unfixed, known or unknown, asserted or unasserted, choate or inchoate, liquidated or unliquidated, secured or unsecured.
- (e) <u>Litigation</u>. To Coherent's Knowledge, there is no action, suit, proceeding, or investigation pending or currently threatened against Coherent, or another third party, affecting or which could affect the Transferred Assets, Licenses or the Business including, but not limited to, any of the foregoing that questions the validity of this Agreement, or the right of Coherent to enter into this Agreement, or to consummate the transactions contemplated hereby, or that might have a material adverse effect, either individually or in the aggregate, on the Transferred Assets or the Licenses.
- (f) <u>Title to Properties</u>; <u>Absence of Liens</u>. Coherent has good and, valid and marketable title to all of the Transferred Assets, which it purports to own free and clear of all Liens, except for Liens that are immaterial in character, amount, and extent, and which do not detract from the value or interfere with the present or proposed use of the properties they affect ("**Permitted Liens**").
 - (g) No Notice of Infringement of Third Parties' Intellectual Property.
 - (i) Coherent has no Knowledge of and has not received notice from any third party that Coherent Lasers, or Coherent's use of Licensed IPR or practice under Licensed Patents infringes any third party patents;
 - (ii) Coherent has no Knowledge of and has received no notice from a third party that the Licensed Patents are invalid or unenforceable;

- (iv) Coherent has no legal opinions pertaining to the infringement of any third-party patents by the Business or the invalidity or unenforceability of the Licensed Patents; and
- (v) Coherent has no Knowledge that Coherent Lasers, Licensed IPR or Licensed Patents violate any third party's trade secret rights.

(h) Taxes.

- (i) To Coherent's Knowledge, no affirmative agreement, consent or election for federal, state, local or foreign Tax purposes has been filed or entered into which would adversely affect or be binding upon the Transferred Assets or the Business after the Effective Date.
- (i) <u>Knowledge</u>. As used in this Article, "**Knowledge**" with respect to Coherent means the actual knowledge of the individuals listed on Schedule 7.1(i).
- (j) <u>Licenses</u>. The Licensed Patents represent all of the patents owned, licensed or controlled by Coherent (i) necessary to manufacture, use, import, sell or have made for sale the Transferred Lasers, (ii) necessary to practice the Licensed IPR, or (iii) otherwise related to the Transferred Assets.
- (k) EXCEPT AS EXPRESSLY SET FORTH IN THIS ARTICLE 7, COHERENT MAKES NO REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, IN RESPECT OF ANY OF ITS ASSETS, LIABILITIES OR OPERATIONS (INCLUDING THE TRANSFERRED ASSETS, ANY COHERENT INTELLECTUAL PROPERTY RIGHTS, AND ANY OTHER THINGS OR RIGHTS PROVIDED, SOLD, TRANSFERRED OR LICENSED BY COHERENT TO LUNA HEREUNDER), INCLUDING WITH RESPECT TO MERCHANTABILITY, NON-INFRINGEMENT OR FITNESS FOR ANY PARTICULAR PURPOSE, AND ANY SUCH OTHER REPRESENTATIONS OR WARRANTIES ARE HEREBY EXPRESSLY DISCLAIMED. COHERENT MAKES NO REPRESENTATION OR WARRANTY AS TO WHETHER THE TRANSFERRED ASSETS (OR ANY COHERENT INTELLECTUAL PROPERTY RIGHTS AND OTHER THINGS OR RIGHTS PROVIDED, SOLD, TRANSFERRED OR LICENSED BY COHERENT TO LUNA HEREUNDER) CONSTITUTE ALL OF THE ASSETS NECESSARY OR SUFFICIENT FOR THE CONTINUED MANUFACTURE OF THE LASERS AND COHERENT MAKES NO REPRESENTATION OR WARRANTY WITH RESPECT TO ANY FORECASTS, FINANCIAL OR OTHERWISE. LUNA HEREBY ACKNOWLEDGES AND AGREES THAT, EXCEPT TO THE EXTENT SPECIFICALLY SET FORTH IN THIS ARTICLE 7, LUNA IS PURCHASING THE TRANSFERRED ASSETS ON AN "AS-IS, WHERE-IS" BASIS. WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, COHERENT MAKES NO REPRESENTATION OR WARRANTY REGARDING ANY ASSETS OTHER THAN THE TRANSFERRED ASSETS AND NONE SHALL BE IMPLIED AT LAW OR IN EQUITY. TO THE EXTENT ANY PRODUCTS ARE INCLUDED WITHIN THE TRANSFERRED ASSETS), THE

WARRANTY PROVIDED WITH RESPECT TO SUCH PRODUCTS IS SOLELY AND EXCLUSIVELY AS SET FORTH IN THIS ARTICLE 7, EVEN IF COHERENT WOULD NORMALLY PROVIDE SUCH PRODUCTS UNDER A SEPARATE WARRANTY.

- 7.2 Representations and Warranties of Luna. Luna hereby represents and warrants to Coherent as of the date of this Agreement as follows:
- (a) <u>Organization, Good Standing and Qualification</u>. Luna is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware.
- (b) <u>Authority</u>. Luna has all requisite corporate power and authority to enter into this Agreement and to perform its respective obligations and consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary action (corporate or otherwise) on the part of Luna, and no further action is required on the part of Luna to authorize the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby. This Agreement has been duly executed and delivered by Luna and is valid and enforceable against Luna in accordance with its terms.
- (c) No Conflict. Neither the execution and delivery of this Agreement, nor the consummation of the transactions contemplated hereby, conflicts or will conflict with, or result in any violation of, or default under (with or without notice or lapse of time, or both), or give rise to a conflict under (i) any provision of Luna's charter documents or bylaws, each as currently in effect, (ii) any contract to which Luna is subject, or (iii) any judgment, order, writ, injunction, decree, award, statute, law, ordinance, rule or regulation applicable to Luna or its properties or assets. No authorization, approval or consent of, and no registration or filing with, any governmental or regulatory official, body or authority is required in connection with the execution, delivery or performance of this Agreement by Luna.

8. CONDITIONS CONCURRENT

8.1 Effectiveness of Luna's Obligations

The effectiveness of Luna's obligations hereunder is conditioned upon Luna's receipt of:

(a) such bills of sale with covenants of warranty, assignments, endorsements, and other good and sufficient instruments and documents of conveyance and transfer, in form reasonably satisfactory to Luna and its counsel, as shall be necessary and effective to transfer and assign to, and vest in, Luna all of Coherent's right, title and interest in and to the Transferred Assets, including without limitation, (A) good and valid title in and to all of the Transferred Assets owned by Coherent, (B) good and valid leasehold interests in and to all of the Transferred Assets leased by Coherent as lessee, if applicable, and (C) all of Coherent's rights relating to the Transferred Assets under all agreements, contracts, commitments, leases, plans, bids, quotations, proposals, instruments and other documents included in the Transferred Assets to which Coherent is a party or by which it has rights on the Effective Date;

(b) all of the Documentation; and

(c) such other documents to be delivered by Coherent hereunder as are reasonably necessary for Luna to effectuate and document the transactions contemplated hereby.

8.2 Effectiveness of Coherent's Obligations

The effectiveness of Coherent's obligations hereunder is conditions upon Luna's payment of the First Payment as set forth in Section 2.6 and upon Coherent's receipt of such other documents to be delivered by Luna hereunder as are reasonably necessary for Coherent to effectuate and document the transactions contemplated hereby.

9. OTHER COVENANTS AND AGREEMENTS

9.1 <u>Additional Documents and Further Assurance</u>. Subject to the terms and conditions herein provided, each Party, at the request of the other, shall execute and deliver such instruments, deeds, certificates, bills of sale, assignments, transfers, conveyances, powers of attorney, assurances and other documents as may be reasonably requested by the other Party and do and perform such other acts and things as may be necessary, proper or desirable, consistent with all applicable laws, rules and regulations, for effecting completely the consummation of this Agreement and the transactions contemplated hereby.

9.2 Confidential Information.

- (a) "Confidential Information" means any information: (i) disclosed by one Party (the "Disclosing Party") to the other (the "Receiving Party"), which, if in written, graphic, machine-readable or other tangible form is marked as "Confidential" or "Proprietary," or which, if disclosed orally or by demonstration, is identified at the time of initial disclosure as confidential and such identification is reduced to a writing delivered to the Receiving Party within thirty (30) days of such disclosure; or (ii) which is otherwise deemed to be "Confidential" by the terms of this Agreement. The Confidential Information of Coherent will include without limitation any information embodied in the Transferred Assets, whether or not so marked.
- (b) <u>Confidential Information Exclusions</u>. Confidential Information will exclude information that the Receiving Party can demonstrate: (i) was independently developed by the Receiving Party without any use of the Disclosing Party's Confidential Information or by the Receiving Party's employees or other agents (or independent contractors hired by the Receiving Party) who have not been exposed to the Disclosing Party's Confidential Information; (ii) becomes known to the Receiving Party, without restriction, from a source (other than the Disclosing Party) that had a right to disclose it without breach of this Agreement; (iii) was in the public domain at the time it was disclosed or enters the public domain through no act or omission of the Receiving Party; or (iv) was rightfully known to the Receiving Party, without restriction, at the time of disclosure.
- (c) <u>Confidentiality Obligations</u>. The Receiving Party shall treat as confidential all of the Disclosing Party's Confidential Information and shall not use such Confidential Information except as expressly permitted under this Agreement. Without limiting the foregoing, the Receiving Party shall use at least the same degree of care that it uses to prevent the disclosure of its own confidential information of like importance, but in no event with less than reasonable care, to prevent the disclosure of the Disclosing Party's Confidential Information.

Subject to the foregoing, Luna will have the right to use all Coherent Confidential Information in accordance with the Licenses to the Transferred Technology granted to Luna pursuant to this Agreement.

- (d) <u>Confidentiality of Agreement</u>. Each Party agrees that the terms and conditions, but not the existence, of this Agreement will be treated as the other Party's Confidential Information and may not be disclosed without the prior written consent of the other Party; <u>provided</u>, <u>however</u>, that each Party may disclose the terms and conditions of this Agreement: (i) in connection with the requirements of an initial public offering or securities filing; (ii) in confidence, to accountants, banks, and financing sources and their advisors; (iii) in confidence, in connection with the enforcement of this Agreement or rights under this Agreement; or (iv) in confidence, in connection with a merger or acquisition or proposed merger or acquisition, or the like.
- (e) <u>Remedies</u>. Unauthorized use by a Party of the other Party's Confidential Information will diminish the value of such information. Therefore, if a Party breaches any of its obligations with respect to confidentiality or use of Confidential Information hereunder, the other Party will be entitled to seek equitable relief to protect its interest therein, including injunctive relief, as well as money damages.
- (f) No Confidential Information of Other Parties. Each Party represents and warrants to the other that it has not used and shall not use in the course of its performance hereunder, and shall not disclose to the other, any confidential information of any third party, unless it is expressly authorized in writing by such third party to do so.
- (g) <u>Required Disclosures</u>. In the event the Receiving Party is required to disclose the Disclosing Party's Confidential Information pursuant to the order or requirement of a court, administrative agency, or other governmental body, the Receiving Party shall provide prompt notice thereof to the Disclosing Party and shall use its reasonable efforts to obtain a protective order or otherwise prevent public disclosure of such information.
- (h) <u>Public Announcements</u>. Neither Luna nor Coherent shall make any public announcement relating to this Agreement until both Luna and Coherent approve in writing the timing, form and content of a public announcement, such approval not to be unreasonably withheld or delayed.
- 9.3 <u>Insurance</u>. Coherent shall add Luna as an additional insured party to its property insurance policies under which the Transferred Assets are insured for the period beginning on the Effective Date and ending not less than five (5) Business Days following the Training Completion Date.
- 9.4 <u>Cooperation in Litigation</u>. The Parties shall reasonably cooperate with each other, at the requesting Party's expense, in the prosecution or defense of any dispute or litigation or other proceeding arising from the Licenses or their respective ownership of the Transferred Assets and involving one or more third parties, including but not limited to affording reasonable access to employees and access to, and information regarding, amounts in dispute and documentation created with respect to the Licenses or the Transferred Assets relating to such dispute or litigation.

9.5 Cooperation Regarding Training Obligations

Coherent shall reasonably cooperate with Luna and the Luna Technical Personnel in order to satisfy the requirements related to the Training Obligations set forth in Section 4.1 on or prior to the Training Completion Date.

10. LIMITATIONS OF LIABILITY

- 10.1 <u>Total Liability</u>. EACH PARTY'S TOTAL LIABILTY TO THE OTHER PARTY HEREUNDER FOR ANY KIND OF LOSS OR DAMAGE ARISING UNDER OR IN CONNECTION WITH THIS AGREEMENT, UNDER ANY THEORY OF LIABILTY (EXCEPT FOR FRAUD), WILL BE LIMITED TO THE AGGREGATE AMOUNT PAID BY LUNA TO COHERENT UNDER THIS AGREEMENT.
- 10.2 EXCLUSION OF DAMAGES. EXCEPT FOR FRAUD, IN NO EVENT WILL EITHER PARTY BE LIABLE TO THE OTHER FOR LOST PROFITS, COST OF PROCUREMENT OR SUBSTITUTE GOODS, OR FOR ANY INDIRECT, SPECIAL, RELIANCE, INCIDENTAL OR CONSEQUENTIAL DAMAGES, WHETHER BASED ON BREACH OF CONTRACT, TORT (INCLUDING NEGLIGENCE), PRODUCT LIABILITY, OR OTHERWISE, AND WHETHER OR NOT THE OTHER PARTY HAS BEEN ADVISED OF THE POSSIBLITY OF SUCH DAMAGES.
- 10.3 <u>Failure of Essential Purpose</u>. The limitations specified in this Article 10 will survive and apply even if any limited remedy specified in this Agreement is found to have failed its essential purpose. LUNA EXPRESSLY AGREES THAT THE PURCHASE PRICE AND ANY ROYALTY PAYMENTS TO COHERENT HEREUNDER REFLECT THE LIMITATIONS OF ANY WARRANTIES AND LIABILITIES OF COHERENT HEREUNDER, AND THAT, BUT FOR SUCH LIMITATIONS, COHERENT WOULD NOT HAVE ENTERED INTO THIS AGREEMENT.

11. TERM AND TERMINATION

- 11.1 **Term**. The term of this Agreement will begin on the Effective Date and will continue until the end of the tenth (10th) Contract Year; *provided, however*, that the representations and warranties in Section 7.1 shall only survive for a period of two (2) years after the Effective Date, unless this Agreement is earlier terminated pursuant to Section 11.2. Notwithstanding the forgoing, the following Sections of this Agreement shall survive beyond the tenth (10th) Contract Year unless otherwise terminated in accordance with their terms or terminated in accordance with Section 11.2: Sections 1.1, 1.2, 3.1 through 3.5, 9.2, 9.4, 10.1 through 10.3, and 12.1 through 12.18.
 - $11.2 \; \underline{\textbf{Termination}}$. This Agreement and all Licenses hereunder may be terminated:
 - (a) By mutual written consent of Coherent and Luna;

- (b) By Coherent if Luna is in breach of any material term of this Agreement (including, without limitation, license and payment terms), and Luna fails to cure the breach within thirty (30) days after receiving written notice of the breach from Coherent;
- (c) By Luna if Coherent is in breach of any material term of this Agreement (including, without limitation, representations and warranties), and Coherent fails to cure the breach within (30) days after receiving written notice of the breach from Luna;
- (d) By Luna (i) at any time after the Effective Date if a court of competent jurisdiction issues a final, non-appealable order enjoining Luna from making, having made, importing, selling or offering for sale Lasers, or from practicing the Licensed Patents or Licensed IPR, or (ii) at any time after the Full Payment Date at Luna's convenience in its sole discretion; or
- (e) By any Party if a Governmental Authority shall have issued an order, decree or ruling or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the transactions contemplated hereby, which order, decree, ruling or other action is final and nonappealable.
- 11.3 **Notice of Termination**. Any termination of this Agreement under Section 11.2 will be effective immediately upon the delivery of written notice thereof by the terminating Party to the other parties hereto (or, in the case of termination pursuant to Section 11.2(b) or Section 11.2(c), on the date specified therein).
- 11.4 <u>Effect of Termination</u>. In the event of termination of this Agreement as provided in Section 11.2, this Agreement shall forthwith become void and there shall be no liability or obligation on the part of any Party hereto, or its affiliates, officers, directors or stockholders, provided that each Party shall remain liable for any breaches of this Agreement prior to its termination. Notwithstanding the foregoing, no termination of this Agreement shall relieve any Party from liability for any breach hereof prior to such termination.

12. GENERAL

- 12.1 Force Majeure. Except with respect to obligations related to the payment of money, neither Party will be held responsible for any delay or failure in performance hereunder to the extent caused by fire, strike, flood, embargo, labor dispute, delay or failure of any subcontract hereof, act of sabotage, riot, accident, delay of carrier, voluntary or mandatory compliance with any governmental act, regulation or request, act of God or by public enemy, or any act or omission or other cause beyond such Party's commercially reasonable control. If any such contingency shall occur, the time for such performance will be deemed extended by the length of time such contingency continues.
- 12.2 <u>No Agency</u>. Except as expressly provided herein, each Party shall in all matters relating to this Agreement act as an independent contractor. Neither Party will have authority, nor will either Party represent that it has any authority, to assume or create any obligation, express or implied, on behalf of the other, or to represent the other Party as agent or employee or in any other capacity. Neither execution nor performance of this Agreement will be construed to have established any agency, joint venture, or partnership.

- 12.3 <u>Fees and Expenses</u>. Each Party will bear its own fees and expenses in connection with this Agreement and the transactions contemplated hereby.
- 12.4 <u>Attorneys' Fees</u>. If any legal action or other legal proceeding relating to this Agreement or the enforcement of any provision of this Agreement is brought against any Party to this Agreement, each side shall bear its own costs and expenses, including attorneys fees and costs.
- 12.5 Notices. Any notice or other communication required or permitted to be delivered to any Party under this Agreement must be in writing and will be deemed properly delivered, given and received when delivered (by hand, by registered mail, by courier or express delivery service or by facsimile) to the address or facsimile telephone number set forth beneath the name of such Party below (or to such other address or facsimile telephone number as such Party may have specified in a written notice given to the other Party):

if to Coherent: 5100 Patrick Henry Drive

Santa Clara, California 95054 Attention: General Counsel Telephone: (408) 764-4180 Facsimile No.: (408) 970-9998

with a copy to: Wilson Sonsini Goodrich & Rosati

Professional Corporation 650 Page Mill Road Palo Alto, California 94304

Attention: Selwyn B. Goldberg, Esq.

Telephone: (650) 493-9300 Facsimile No. (650) 496-4006

if to Luna: 1703 S. Jefferson Street, SW

Suite 400

Roanoke, Virginia 24016 Attention: General Counsel Telephone: (540) 769-8400 Facsimile No. (540) 581-0951

with a copy to: Wilson Sonsini Goodrich & Rosati

Professional Corporation

Two Fountain Square, Reston Town Center

11921 Freedom Drive, Suite 600

Reston, Virginia 20190

Attention: Trevor J. Chaplick, Esq. Telephone: (703) 734-3100 Facsimile No. (703) 734-3199

 $12.6 \, \underline{\text{Governing Law}}$. THIS AGREEMENT WILL BE GOVERNED IN ALL RESPECTS BY THE LAWS OF THE UNITED STATES OF AMERICA AND THE STATE

OF DELAWARE WITHOUT REGARDS TO CONFLICTS-OF-LAWS PRINCIPLES. The rights and obligations of the Parties under this Agreement will not be governed by the 1980 U.N. Convention on Contracts for the International Sale of Goods.

- 12.7 <u>Injunctive Relief</u>. It is understood and agreed that, notwithstanding any other provision of this Agreement either Party's breach of confidentiality obligations or provisions relating to proprietary rights will cause irreparable damage for which recovery of money damages would be inadequate, and that the other Party will therefore be entitled to seek timely, injunctive relief to protect such Party's rights under this Agreement in addition to any and all remedies available at law.
- 12.8 <u>Waiver</u>. (a) No failure on the part of a Party to exercise any power, right, privilege, or remedy under this Agreement, and no delay on the part of any Party in exercising any power, right, privilege, or remedy under this Agreement, will operate as a waiver of such power, right, privilege, or remedy; and no single or partial exercise of any such power, right, privilege, or remedy will preclude any other or further exercise thereof or of any other power, right, privilege, or remedy.
- (b) No Party will be deemed to have waived any claim arising from this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such Party; and any such waiver will not be applicable or have any effect except in the specific instance in which it is given.
- 12.9 <u>Assignment</u>. Either Party may assign, directly or indirectly, this Agreement, or any of its rights or obligations hereunder, to any third party at any time. Subject to the foregoing, this Agreement shall inure to the benefit or the Parties, their successors and permitted assigns.
- 12.10 <u>Severability</u>. If, for any reason, a court of competent jurisdiction finds any provision of this Agreement, or portion thereof, to be invalid or unenforceable, such provision of the Agreement will be enforced to the maximum extent permissible so as to effect the intent of the Parties, and the remainder of this Agreement will continue in full force and effect. The Parties agree to negotiate in good faith an enforceable substitute provision for any unenforceable provision that most nearly achieves the intent and economic effect of the unenforceable provision. Notwithstanding the foregoing, if a court of competent jurisdiction determines that any restriction on any license granted herein is invalid or unenforceable, then the license grants to which such restriction relates shall terminate automatically.
- 12.11 <u>Entire Agreement</u>. This Agreement and the Exhibits and Schedules hereto set forth the entire understanding of the Parties hereto relating to the subject matter hereof and supersede all prior agreements and understandings between the Parties hereto relating to the subject matter hereof.
- 12.12 <u>Amendments</u>. This Agreement may not be amended, modified, altered or supplemented other than by means of a written instrument duly executed and delivered by representatives of both of the Parties.

- 12.13 Exports. Each of the Parties acknowledges that the other Party was, is or may be subject to regulation by agencies of the U.S. Government, including, but not limited to, the U.S. Department of Commerce, which prohibit export or diversion of certain products and technology to certain countries. Any and all obligations of either Party to provide products, documentation, or any media in which any of the foregoing is contained, as well as any other technical assistance will be subject to such United States laws and regulations as shall from time to time govern the license and delivery of technology and products abroad by persons subject to the jurisdiction of the United States, including the Export Administration Act of 1979, as amended, any successor legislation, and the Export Administration Regulations issued by the Department of Commerce, Bureau of Export Administration. Each of the Parties shall use commercially reasonable efforts to cooperate with the other Party, with any resulting expenses of the cooperating Party to be paid or reimbursed by the other Party, including without limitation providing required documentation, in order to obtain export licenses or exemptions therefrom. Each of the Parties warrants that it will comply with the Export Administration Regulations and other United States laws and regulations governing exports in effect from time to time, and any export laws and regulations of any other country to which such Party may ship any of the Transferred Assets or Lasers made using the Transferred Assets.
- 12.14 **No Third Party Rights**. This Agreement is not intended and shall not be construed to create any rights in any parties other than Coherent and Luna and no other person shall assert any rights as a third party beneficiary hereunder.
- 12.15 Exhibits and Schedules. The Exhibits and Schedules attached hereto are incorporated into this Agreement and shall be deemed a part hereof as if set forth herein in full. References herein to "this Agreement" and the words "herein," "hereof" and words of similar import refer to this Agreement (including Exhibits and Schedules) as an entirety. In the event of any conflict between the provisions of this Agreement and any such Exhibit or Schedule, the provisions of this Agreement shall control.
- 12.16 <u>Public Announcements</u>. Neither Coherent, Luna, their respective affiliates, nor the agents and representatives of Coherent, Luna or their affiliates will make any press release or public announcement concerning the existence of this Agreement or the transactions contemplated hereby, except to the extent legally required to do so after consultation with the other Party hereto, or with the prior consent of the other Party hereto.

12.17 Acknowledgement

The Parties each acknowledge that all the terms and conditions in this Agreement have been the subject of active and complete negotiation between the Parties and represent the Parties' agreement based upon all relevant considerations. The Parties agree that the terms and conditions of this Agreement shall not be construed in favor of or against any Party by reason of the extent to which any Party or its professional advisors participated in the preparation hereof or thereof.

12.18 Counterparts. This Agreement may be executed in counterparts, which, when taken together, shall constitute one and the same agreement.

IN WITNESS WHEREOF, the Parties, by their duly authorized representatives, have executed this Agreement as of the Effective Date.

Coherent, Inc.

Luna Innovations Incorporated

By: /s/ John Ambroseo

Name: John Ambroseo

Title: CEO

By: /s/ Scott A. Graeff

Name: Scott A. Graeff

Title: Chief Commercialization Officer

SCHEDULES

Schedule 1.1(e)

Clean Room Schedule

Schedule 1.1(bb)

Licensed Patents

Schedule 1.1(tt)

Bill of Material for Transferred Lasers

Schedule 2.1

Transferred Assets

Schedule 2.2

Excluded Assets

All tools listed in Clean Room Schedule (Schedule 1.1(e)) that are not identified as Luna transfer tools.

Schedule 3.3

Prior License Restrictions

None.

Schedule 7.1(i)

John Ambroseo
Mark Gitin
Hal Jerman

Joe Drake

Scott Miller

Russell Austin